Author’s response to reviews

Title: Predictors of mortality among HIV Infected Children on Anti-Retroviral Therapy in Mekelle Hospital, Northern Ethiopia, A retrospective cohort study

Authors:

Aregay G G/medhine AG (aregaygebremedhin@gmail.com)
Solomon G G/mariam SG (solomongmariam@yahoo.com)
Carla D Decotelli CD (carladecotelli@gmail.com)
Berhe S W/aregawi BW (berheph@yahoo.com)
Fisaha T Haile FH (fisaha.1999@yahoo.com)

Version: 5 Date: 7 December 2012

Author’s response to reviews:

Cover letter

We the authors would like to publish our article entitled as “Predictors of mortality among HIV Infected Children on Anti-Retroviral Therapy (ART) in Mekelle Hospital, Northern Ethiopia, A retrospective cohort study” in your journal because BMC Public Health pen access policy allows maximum visibility of articles published in the journal as they are available to a wide, global audience. More over, BMC Pregnancy and Childbirth offers a fast publication schedule; online manuscript submission and peer review is fully managed electronically.

We the authors would like to confirm that the work presented here is our own original work except where otherwise indicated. We would also like to assure you that this work has ethically cleared and has no competing interests with the funding organizations. More over, the manuscript has not submitted to any journal nor considered for publication by any journal. Finally, the manuscript has been read and approved by all authors indicated here.